Author's response to reviews

Title: The importance of central corneal thickness measurements and decision making in general ophthalmology clinics: a masked observational study

Authors:

Ashish A Patwardhan (apatwa@hotmail.com)
Mohammad Khan (drmikhan@gmail.com)
Susan P Mollan (soozmollan@doctors.org.uk)
Paul Haigh (paul.haigh@rsh.nhs.uk)

Version: 7 Date: 16 January 2008

Author's response to reviews: see over
To
The Editor
BMC Ophthalmology
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB
UK

Dear Sir/Madam

I have attached the final revised manuscript incorporating formatting changes as suggested. With this letter I have attached point-by-point reply to the suggested changes. I was not sure how you want me write affiliations on the title page but I hope I have got it right! Thanks

Yours sincerely

Ashish Patwardhan
1. Affiliations: Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution. - Changes made

2. Please delete the competing interests section from the title page and please change the statement in the competing interests section above the authors contributions to 'The authors declare that they have no competing interests.' - Changes made

3. Please crop the excess white space from around figure 2. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures - Changes made

4. We could not find reference 16 cited within the text. Please ensure that reference 16 is cited between references 15 and 17 in the text. Please update the text so that the references are mentioned in order - Changes made